Q: SPECIFIC COMMENTS TO AUTHORS
The occurrence and development of gastric cancer is a multistage process involving changes at the gene and molecular levels. There is a period of precancerous lesions in the early stage of gastric cancer, and most of the precancerous lesions remain unchanged, while some develop into cancer. Most gastrointestinal malignant tumors have no obvious symptoms during the early stage, they cannot be detected quickly. Early diagnosis and early treatment of gastric cancer is the key to improving the survival and cure rates of patients. As one of the most common tumor markers, the cell surface structural antigen CEA is widely used as a diagnostic and monitoring index for various gastrointestinal tumors. In this study, the authors examined 13 tumor-related indices, 38 clinical biochemical indices and 10 cytokines in gastric cancer and gastric polyp patients. The idea of the study is interesting, and the study is well display. Methods are described in detail, and results are very informative. Some minor editing is required before acceptance. Please take attention about some minor language polishing, and update the reference according to the guideline of the journal.

A: Thanks. We have carefully checked and revised the language, and we also updated the reference according to the guideline of the journal.
Q: SPECIFIC COMMENTS TO AUTHORS
This is a very interesting study of the blood index panel for gastric cancer detection. This study is very well designed and the panel of this study may become a detection method for clinical practice. Tumor-related and clinical biochemical index detection, cytokine detection in this study is well described. The results data are interesting, and well discussed. The reviewer recommends to accept this study after a minor editing. Thank you for inviting me to review this interesting study.
A: Thanks for your comments on our study.